Author's response to reviews

Title: Sutureless Management of Left Ventricle Wall Rupture; a Series of Three Cases

Authors:

Remco Bergman (r.bergman@umcg.nl)
Jayant S Jainandunsing (j.s.jainandunsing@umcg.nl)
Bozena Woltersom (b.d.woltersom@umcg.nl)
Inez J Den Hamer (i.j.den.hamer@umcg.nl)
Ehsan Natour (e.natour@umcg.nl)

Version: 2 Date: 28 April 2014

Author's response to reviews:

1. Abstract section on page 2 has been improved. It does not exceed 350 words. And contains an introduction about why this case is important and needs to be reported. Brief details of what the patient(s) presented with, including the patient’s age, sex and ethnic background has also been included. A brief conclusion of what the reader should learn from the case report has been added.

2. Keywords representing the main content of the article, have been added

3. Case presentations includes brief details of what the patient(s) presented with, including the patient's age, sex and ethnic background.

4. the Ethnicity of the patients in the Abstract and Case presentation sections, is reported now

5. Please include a Competing Interests section before the Reference list, has been added

6. A list of abbreviations used in the manuscript and their meanings after the Conclusions section, has been added.

7. An acknowledgement section at the end of the manuscript before the reference list, has been added.